Author's response to reviews

Title: Expression of RNA interference triggers from an oncolytic herpes simplex virus results in specific silencing in tumour cells in vitro and tumours in vivo.

Authors:

Marianna Anesti (MAnesti@biovex.com)
Guy R Simpson (g.simpson@surrey.ac.uk)
Toby Price (TPrice@biovex.com)
Hardev S Pandha (h.pandha@surrey.ac.uk)
Robert S Coffin (rcoffin@biovex.com)

Version: 4 Date: 10 September 2010

Author's response to reviews: see over
8\textsuperscript{th} September 2010

Dr Guy Simpson
Oncology
Postgraduate Medical School
Daphne Jackson Road
University of Surrey
Guildford, GU2 7WG
08.09.2010

Dear Editor

MS:1897727126336065 - Expression of RNA interference triggers from oncolytic herpes simplex virus results in specific silencing in tumour cells in vitro and tumours in vivo

We have made the changes to our manuscript that you requested. They are listed below:

Major revisions (we require the author to make these changes)

1) Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is ¿First name Initial Last name¿ (e.g. Joe F Bloggs. We have altered the authors¿ list to comply with style.

2) Authors' contributions: Please indicate that all authors read and approved the final manuscript. We have added this statement to the authors' contributions section.

3) References: please cite all references in numerical order in the text. References 55-69 have not been cited. We have changed to square brackets

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above

4) Track changes: please remove. – removed

5) Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance. Checked.

Best regards,

Guy R Simpson
Anna-Maria Anesti